Author's response to reviews

Title: Acromegaly resolution after traumatic brain injury: case report

Authors:

Alejandro Cob (alejandro.cobg@gmail.com)

Version: 2 Date: 20 February 2014

Author's response to reviews: see over
Dear Editor-in-Chief,

I am submitting to you my manuscript entitled “Acromegaly resolution after traumatic brain injury”. I would like to have this manuscript considered for publication as a Case Report in the Journal of Medical Case Reports.

To my knowledge, this is the first report showing a patient with acromegaly who resolves his condition after a traumatic brain injury. Post-traumatic hypopituitarism is a well known condition in patients with previous normal pituitary function, but an association between a head trauma and resolution of hyperfunctioning pituitary tumors has not been described.

Complete cure of acromegaly is a difficult goal to achieve, even after surgery and medical treatment. I believe this case would be of interest for the specialties of endocrinology and neurosurgery because it describes an uncommon cause of growth hormone hypersecretion resolution and tumor disappearance.

I confirm that this manuscript has not been published elsewhere and is not under consideration by another journal.

I am the only author of the manuscript and I am determined with its submission to the Journal of Medical Case Reports.

Please let me know of your decision at your earliest convenience.

With my best regards,

Alejandro Cob, MD

Please address all correspondence to: alejandro.cobg@gmail.com